Author's response to reviews

Title: Green tea polyphenols and Tai Chi for bone health: Designing a placebo-controlled randomized trial

Authors:

Chwan-Li Shen (leslie.shen@ttuhsc.edu)
Ming-Chien Chyu (m.chyu@ttu.edu)
James K Yeh (jyeh@winthrop.org)
Carol K Felton (carol.felton@ttuhsc.edu)
Ke T Xu (ke.xu@ttuhsc.edu)
Barbara C Pence (barbara.pence@ttuhsc.edu)
Jia-Sheng Wang (jswang@uga.edu)

Version: 2 Date: 27 August 2009

Author's response to reviews: see over
Re: Green tea polyphenols and Tai Chi for bone health: Designing a placebo-controlled randomized trial

Dear Editor-In-Chief:

Thank you so much for accepting of our manuscript. We have made the changes accordingly including the following:
(i) provision of completed affiliations,
(ii) provision completed citation for reference 15,
(iii) removal of figure title and addition of figure legend after reference list,
(iv) removal of line number,
(v) minimizing figure space.

If you have any questions, please let me know.

Sincerely yours,

Chwan-Li (Leslie) Shen, Ph.D.
Associate Professor of Pathology
School of Medicine
Texas Tech University Health Sciences Center
Lubbock, TX 79430-9097, USA
Tel: 806-743-2815
Fax: 806-743-2766
e-Mail: Leslie.Shen@ttuhsc.edu